Author’s response to reviews

Title: Type 1 diabetes-associated cognitive impairment and diabetic peripheral neuropathy in Chinese adults: results from a prospective cross-sectional study

Authors:

Xin Ding (dingxin@suda.edu.cn)

Chen Fang (afa9911@sina.com)

Xiang Li (lx_zj@163.com)

Yongjun Cao (yongjuncao@126.com)

Qilin Zhang (zhangqilin_1@163.com)

Yun Huang (hy0625@126.com)

Jian Pan (panjian2008@163.com)

Xia Zhang (stefence1025@163.com)

Version: 1 Date: 02 Jul 2018

Author’s response to reviews:

Dear editors,

1. If improvements to the English language within your manuscript have been requested, you should have your manuscript reviewed by someone who is fluent in English.

Ans: Thanks for the valuable advice. I have my manuscript reviewed by someone who is fluent in English.

2. In accordance with BioMed Central editorial policies and formatting guidelines, all submissions to BMC Endocrine Disorders must have a Declarations section which includes the mandatory sub-sections listed below. Please refer to the journal’s Submission Guidelines web page for information regarding the criteria for each sub-section.

Ans: We appreciate the editors’suggestions. I have checked every sub-section according to the journal’s Submission Guidelines and improved my Declarations section in my manuscript. I have added the sub-section of Authors’ Information. And for the 'Availability of data and materials' section, the data and materials have been available from the corresponding author on reasonable request as stated in the Declarations section of my manuscript.
Yours sincerely,

Xia Zhang